Author’s response to reviews

Title: Situational Awareness within Objective Structured Clinical Examination (OSCE) Stations in Undergraduate Medical Training - A Literature Search

Authors:

Markus Fischer (m.fischer2@nuigalway.ie)
Kieran Kennedy (kieran.kennedy@nuigalway.ie)
Jean Ker (j.s.ker@dundee.ac.uk)
Paul O’Connor (paul.oconnor@nuigalway.ie)
Eva Doherty (edoherty@rcsi.ie)
Thomas Kropmans (thomas.kropmans@nuigalway.ie)

Version: 4 Date: 07 Dec 2017

Author’s response to reviews:

Galway, 7th of December 2017

Dear Editor,

Thank you most sincerely for your positive review of this submission. We are most pleased to be working towards publication with BMC Medical Education.

The following is a point-by-point response to your feedback.

1) You requested that we incorporate the information that we had provided as a separate document (i.e. “title page”) in to the main text. That is now completed in our revised submission.
2) You requested that we discard the PRISMA checklist. We have removed that from our revised submission. You requested that we include the PRISMA flowchart as a figure. We have now included that as a figure and we have appropriately adjusted all references to figures within the main body of the text in order to ensure they remain accurate.

3) You requested that we remove all additional files. We had originally included two tables as additional files. We have now incorporated those in to the main text in accordance with the author guidelines. All additional files have now been removed. However, we have submitted our Figures as additional separate files in accordance with author guidelines. If we are mistaken in that regard (i.e. if they also should have been incorporated in to the main manuscript), then please just let us know and we will promptly address same.

4) You requested that we provide Figure legends as a separate section at the end of the manuscript. We have now included that new section in our revised submission.

Yours sincerely,

Markus Fischer
c/o Dr. Thomas Kropmans
National University Ireland, Galway
School of Medicine
Ireland
Email: m.fischer2@nuigalway.ie